Author's response to reviews

Title: A randomised controlled trial of three or one breathing technique training sessions for breathlessness in people with malignant lung disease.

Authors:

Miriam J Johnson (miriam.johnson@hyms.ac.uk)
Mona Kanaan (mona.kanaan@york.ac.uk)
Gerry Richardson (gerry.richardson@york.ac.uk)
Samantha Nabb (s.nabb@hull.ac.uk)
David Torgerson (david.torgerson@york.ac.uk)
Anne English (anneenglish@nhs.net)
Rachael Barton (rachael.barton@hey.nhs.uk)
Booth Sara (sb628@cam.ac.uk)

Version: 5
Date: 12 August 2015

Author's response to reviews:

Dear Prof Alam

Re: A randomised trial of three or one breathing technique training sessions for breathlessness in people with malignant lung disease.
MS: 1671029929172375
Please find the second revision of our manuscript.
We present our responses below in blue italic text.

Yours sincerely

Miriam Johnson, on behalf of the investigator team.

1. please add the abstract, abbreviations and keywords
This has now been done. We have used the option of defining abbreviations in the text when they are first used rather than providing a list of abbreviations.

2. add trial number and registration date at the end of the abstract
This has now been done.

3. the Funding section needs to move into the Acknowledgement section
This has now been done.

4. Proof read to check conformity with Journal style
The references still had volume numbers – these have now been removed